STA 104 MTH 135 Request for Feedback 10:15-10:25 am Tuesday, October 21, 2001 I am interested in your reactions to the course, so far. Please take a few minutes to respond to the following questions. For the Comments it would be helpful for you to mention anything you’d like me to start doing, continue doing, or stop doing.

1) The lectures strike about the right balance between examples & theory:

Definitely NO No maybe Yes Definitely Yes Unsure Comments:

2) There is enough opportunity for help outside the classroom (office hours, web pages, e-mail, daily SECC help sessions in 211 Old Chem, telephone, before/after class):

Definitely NO No maybe Yes Definitely Yes Unsure Comments:

3) Regular quizzes would help me learn the material better:

Definitely NO No maybe Yes Definitely Yes Unsure Comments:

4) The homework assignments are about right in length & difficulty:

Definitely NO No maybe Yes Definitely Yes Unsure Comments:

5) My midterm score was in the range (these are the quintiles, so there are at least 6 students in each range):

0–50 51–67 68–76 77–88 89-100 Unsure Comments:

6) If I could change one thing about the course, it would be: (use back if you need more room, or want to make other comments)